Author's response to reviews

Title: A Forced Titration Study of the Antioxidant and Immunomodulatory Effects of Ambrotose AO Supplement

Authors:

Stephen P Myers (smyers@scu.edu.au)
Lesley Stevenson (LStevenson@hortresearch.co.nz)
Phillip A Cheras (philcheras@yahoo.com.au)
Joan O'Connor (joan.oconnor@scu.edu.au)
Lyndon Brooks (lyndon.brooks@scu.edu.au)
Margaret Rolfe (margaret rolfe@scu.edu.au)
Paul Conellan (paul.conellan@scu.edu.au)
Carol Morris (carol.morris@scu.edu.au)

Version: 3 Date: 25 February 2010

Author's response to reviews:

Dear Editors

All changes both major and minor as requested by Steven Tunney (Editorial Production Assistant) have been made to the manuscript. A conclusion has been added at the end of the discussion without adding further words. Most of the last paragraph was deemed a suitable conclusion and annexed accordingly.

I hope that the manuscript is now ready for publication.

Yours Sincerely

Professor Stephen P Myers